Author's response to reviews

Title: Protocol for a systematic review of prognosis after mild traumatic brain injury: an update of the WHO Collaborating Centre Task Force findings

Authors:

Carol Cancelliere (carol@427health.com)
J David Cassidy (dcassidy@health.sdu.dk)
Pierre Cote (pcote@uhnresearch.ca)
Cesar A Hincapie (cesar.hincapie@utoronto.ca)
Jan Harvigsen (jhartvigsen@health.sdu.dk)
Linda J Carroll (lcarroll@ualberta.ca)
Connie Marras (cmarras@uhnresearch.ca)
Eleanor Boyle (eboyle@uhnresearch.ca)
Vicki Kristman (vkrystma@lakeheadu.ca)
Ryan Hung (rhung@hollandbloorview.ca)
Britt-Marie Stalnacke (brittmarie.stalnacke@rehabmed.umu.se)
Peter Rumney (prumney@hollandbloorview.ca)
Victor Coronado (vgc1@cdc.gov)
Lena W Holm (Lena.Holm@ki.se)
Jorgen Borg (Jorgen.Borg@ki.se)
Catharina Nygren-de Boussard (catharina.nygren-deboussard@ds.se)
Jean-Luc af Geijerstam@sbu.se (geijerstam@sbu.se)
Michelle Keightley (Michelle.Keightley@utoronto.ca)

Version: 2 Date: 12 January 2012

Author's response to reviews: see over
1. The guarantor of the review is J David Cassidy.

2. Some keywords have removed and topic specific MeSH headings have been used.

3. The role of the funding agency has been stated.

4. The process of formulating recommendations has been presented in more detail. The SIGN system will be used.

5. A more general ‘Search’ section has been added.